
LHCb Computing Operations - minutes of the meeting 18th January 2007

Present : Davide Perego, Roberto Santinelli, Nick Brook, Joel Closier, Stuart 
Paterson, Raja Nakadumar, Angelo Carbone, Andrei Tsaregorotsev,  Philippe 
Charpentier, Andrew Smith, Andrew Pickford

Action list:
1 - joel : lhcbdata pool full
2 - joel/stuart : verify that v2r12p6 release is OK for MC and analysis
3 - raja : upgrade the VOBOX to v2r12p6 and restart the transfer agent
4 - Ricardo : check agent cleaner

Simulation : 
Enough jobs in the queue. Run smoothly without any special problem

Reconstruction :
Reconstruction for B-inclusive is running for low and high luminosity. 6 T1 are 
acting at the same time. Big progress. 

VoBox Transfers:
Low transfer rate. It seems that that most of the VoBox transfer agent are 
stuck. Raja will upgrade to the latest version of DIRAC v2r12p6 and restart the 
transfer agent.
It seems that there is some problem of authentification with the LFC.
Raja is working on having NIKHEF VoBox in the game

AOB:
Gridka : Is back in production after the VO_LHCB_SW_DIR has been completely 
cleanup.
SANDBOXSE is the configuration service. To be tested by Stuart.
The agent cleaner is not behaving propely. The "DELETED" jobs are still in the 
monitoring page.
Volhcb02 will be re-installed on Monday.
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